

International Journal of Pharmacology

ISSN 1811-7775





International Journal of Pharmacology

ISSN 1811-7775 DOI: 10.3923/ijp.2019.636.641



Research Article Association of Autoimmune Hypothyroid Disease and Obesity with Vitamin D Deficiency in Female Patients

¹Sadia Batool, ¹Dalia Almaghaslah, ¹Amira Abdel Motaal, ¹Abdulrhman Alsayari, ¹Arwa Khaled, ²Zuha Dirki and ²Shahid Attique

¹College of Pharmacy, King Khalid University, Abha, Saudi Arabia ²Department of Medical, Khamis Mushayt General Hospital, Khamis Mushayt, Saudi Arabia

Abstract

Background and Objective: Vitamin D deficiency is becoming a global problem and is linked to many immune disorders. In regard to autoimmune hypothyroidism, the data is still controversial. This study aimed to evaluate the link between autoimmune hypothyroidism, BMI and vitamin D in the female population of Saudi Arabia. **Materials and Methods:** This was a prospective cross-sectional study. Blood samples for vitamin D (25 OH) and thyroid function (TSH, T3 and T4) were taken from 51 autoimmune hypothyroid female patients (diagnosed after testing positive for autoimmune thyroid antibodies). Vitamin D deficiency was designated at levels lower than 20 ng mL⁻¹ and compared with thyroid function tests. **Results:** The results indicated low levels of vitamin D in up to 90% of these patients. Of the total 51 patients, 5 patients (9.8%) had normal vitamin D levels, 2 patients (3.9%) had insufficient levels and 44 patients (86.3%) had deficient levels. A significant inverse correlation was noted between vitamin D and TSH (r = -0.28, p = 0.046) indicating an increase in TSH levels as vitamin D levels decrease. T3 and T4 were found to be in their normal range. In regard to body mass index (BMI), 10 of the 51 patients (9.6%) had normal body weight, 5 patients (9.8) were overweight and 36 patients (70.6) were obese. The results, however, were statistically non-significant. **Conclusion:** The results indicated that patients with autoimmune hypothyroidism suffered from hypovitaminosis D that was significantly associated with the degree and severity of the disease.

Key words: Vitamin D, immune disorders, body mass index, thyroid profile, autoimmune hypothyroidism

Citation: Sadia Batool, Dalia Almaghaslah, Amira Abdel Motaal, Abdulrhman Alsayari, Arwa Khaled, Zuha Dirki and Shahid Attique, 2019. Association of autoimmune hypothyroid disease and obesity with vitamin D deficiency in female patients. Int. J. Pharmacol., 15: 636-641.

Corresponding Author: Sadia Batool, College of Pharmacy, King Khalid University, Abha, Saudi Arabia Tel: 00966557316186

Copyright: © 2019 Sadia Batool *et al.* This is an open access article distributed under the terms of the creative commons attribution License, which permits unrestricted use, distribution and reproduction in any medium, provided the original author and source are credited.

Competing Interest: The authors have declared that no competing interest exists.

Data Availability: All relevant data are within the paper and its supporting information files.

INTRODUCTION

Vitamin D is an important vitamin, required for the regulation of calcium and bone health. This vitamin increases the uptake of calcium from the gastrointestinal tract (GIT) and links the re-absorption of bone calcium to osteoclast stimulation, thus contributing to bone health. However, evidence has identified other effects as well. Its role in the maintenance of immune function, control of hypertension, certain malignancies and autoimmune diseases was also observed¹.Recently, VDR receptors have been found in various organ systems, including the cells of bone marrow, large intestine, breast and brain². Thus, many tissues can convert inactive vitamin D to an active form, as they carry the enzyme 1 alpha hydroxylase, thus pointing to roles beyond bone regulation³. T-cells, B-cells and antigen presenting cells have also been shown to express VDR receptors and are capable of synthesizing the active form of vitamin D (1, 25, dihydroxy)⁴.

Thyroid dysfunction is considered a common endocrine disease and its prevalence has recently increased⁵. Thyroid problems are divided into hypothyroidism and hyperthyroidism. A defect in the gland could be due to primary or secondary causes⁶. The etiology of autoimmune thyroid disease (ATD) is multifactorial, including genetic, hormonal and environmental factors. Hypothyroidism could also be due to various causes, with an autoimmune disorder being one of the main etiologies⁷. It is characterized by lymphocytic infiltration of the thyroid parenchyma. This disease affects up to 1.5% of the population, predominantly females. The prevalence of hypothyroidism among adult females of different age groups ranges from 3-7.5% and is more frequent among elderly women⁵.

Vitamin D deficiency has been linked to many chronic diseases and autoimmune disorders, such as multiple sclerosis, diabetes mellitus Type 1, rheumatoid arthritis and systemic lupus erythematosus (SLE)⁸⁻¹⁰. However, the relationship between vitamin D and AITD is still a controversial issue¹¹. Many studies have pointed out a correlation between a low concentration of vitamin D and ATD, while other studies have discounted the role of vitamin D in autoimmune hypothyroid disease^{12,13}. Most of these studies were carried out on existing cases of autoimmune hypothyroid disease. This study aimed to evaluate the vitamin D status of newly diagnosed autoimmune hypothyroid female patients, who have not undergone any treatment at the time of presentation. This study also examined the correlation between body mass index (BMI) and vitamin D levels.

MATERIALS AND METHODS

Study design and setting: The study was designed as a single centre, prospective cross-sectional study. It was conducted at the Endocrinology Outpatient Department Clinic, Khamis Mushayt General Hospital, southern region, Kingdom of Saudi Arabia, between October, 2018 and January, 2019. The Research Ethics Committee at College of Medicine, King Khalid University approved the research protocol (REC# 2018-05-37) and consent forms were obtained from participants before the beginning of the study.

Study participants: The sample size was based on a previous study from Qassim, Saudi Arabia¹⁴. The inclusion criteria was as follows: female gender, between the age of 18-45 years, newly diagnosed with autoimmune hypothyroidism and tested positive for ATA (thyroid peroxidase antibody test and anti-thyroglobulin antibody test)¹⁴. The exclusion criteria was as follows: patients with a history of liver or renal disorders, thyroid surgery, primary hyperparathyroidism, medication history that involved taking calcium/vitamin D supplements or drugs that alter the thyroid status, such as amiodarone, lithium or immune modulators. Patients with renal or liver disease. cancer, severe weight loss, gall bladder disease, lipid abnormalities, bowel diseases and those who were pregnant, or were on immunosuppressive medications, insulin, sulfonamides or any supplements including vitamin D were also excluded.

Patients who met the inclusion criteria were invited to participate during their visit to the Endocrinology Outpatient Clinic and were then asked to sign a written informed consent. Blood samples were collected for hematological and chemical analyzed. A total of 51 female patients were included in the study.

Laboratory testing: All biochemical tests were conducted by photometric assays (BT2000) using a Pars kit. TSH was measured by the chemiluminescent immunoassay method (Advia Centaur CP, Siemens Healthcare Diagnostic Inc., USA)¹⁵. Vitamin D levels were measured by an enzyme-linked immunosorbent assay kit (Immunodiagnostic Systems Limited, UK). All samples were taken in a single sitting and without fasting. Blood was drawn from the arm vein using a standard technique. The samples were kept in red top or serum separator vacutainers and stored at -20°C until further analysis.

Data collection form: The data collection form consisted of two sections. Section 1 gathered information on the patient's socio-demographic characteristics, including age, sex, weight, height and nationality. Section 2 collected information regarding laboratory test results, including serum 25 (OH) vitamin D levels and thyroid blood profile (TSH, T3 and T4). The identity of each patient was kept anonymous by assigning codes, P1-P51, to study participants.

Serum TSH, T3 and T4 and Vitamin D (25-OH vitamin D):

Patients' blood was evaluated for thyroid profile and vitamin D levels with the following reference ranges¹⁶:

- TSH: 0.27-4.5 ml U L⁻¹
- T3: 2.8-7 pmoL L⁻¹
- T4: 12-22 pmoL L⁻¹
- Vitamin D (25 OH vitamin D): Normal: >30 ng mL⁻¹, Insufficiency: >20 and <30 ng mL⁻¹, Deficiency: <20 ng mL⁻¹

Body mass index (BMI): Patients BMI was measured as follows; Measures of weight and height were collected in person by the Endocrinology Clinic nurse. Weight measurements were taken on digital scales and heights were measured using portable audiometers. BMI was defined as weight in kilograms divided by height in meters squared. BMI was categorized according to the World Health Organization classifications: underweight (BMI <18.5), normal weight (BMI 18.5-24.9), overweight (BMI 25.0-29.9), class I obese (BMI 30.0-34.9), class II obese (BMI 35.0-39.9) and class III obese (BMI≥40.0).

Statistical analysis: In this research, data was expressed as counts and percentages for categorical variables and as Mean±standard deviation for continuous variables. Data was assessed for normality using the Kolmogorov-Smirnov test and frequencies, means and standard deviations were calculated by descriptive statistics. Correlations between variables were tested using the Pearson or Spearman's correlation analysis. Values between 0-0.3 indicated a weak correlation, 0.3-0.7 indicated intermediate and 0.7-1.0 indicated a strong correlation. A p-value <0.05 was considered statistically significant. Statistical analysis was conducted using the Statistical Package for the Social Sciences (SPSS version 24 for Windows).

RESULTS

A total of 51 female patients were included in the study, almost half of them (25, 49%) were between 40 and 50 years

of age, while 21 (41.2 %) were between 30-39 years and only 5 (9.8%) were between 20-29 years, as shown in Table 1.

Correlation between serum Vitamin D and thyroid profile:

Of the total 51 patients, 5 (9.8%) had normal vitamin D status, 2 (3.9%) were insufficient and 44 (86.3%) patients were deficient (Table 1). The mean and standard deviation were calculated for vitamin D and thyroid blood profile (i.e., T3, T4 and TSH) as summarized in Table 1.

A significant inverse correlation was found between vitamin D and TSH (r = -0.281, p = 0.046) as vitamin D decreased, the level of TSH increased. In this study, participants presented high levels of TSH with a mean of 13 ± 11.1 mL μ L⁻¹ and vitamin D deficiency levels with a mean of 18.8 ± 10.8 ng mL⁻¹. Since all study participants were newly diagnosed with ATD, this finding indicated a possible link between vitamin D deficiency and the occurrence of autoimmune hypothyroidism.

Table 1: Correlation between Serum Vitamin D levels and Thyroid Profile (TSH, T3, T4)

| Demographic characteristics | | Number of p | Number of patients (%) | |
|--|--------------------------|--|------------------------|--|
| Age | | · · | | |
| 20-29 | | 5(9.8%) | 5(9.8%) | |
| 30-39 | | 21(41.2 | 21(41.2%) | |
| 40-50 | | 25(49%) | | |
| Diagnostic criteria | | Number of patients (%) | | |
| Vitamin D serum l | evels | | | |
| Normal: >30 ngmL ^{-1} | | 5 (9.8% | 5 (9.8%) | |
| Insufficient: >20 and <30 ng mL ^{-1} | | 2 (3.9% | 2 (3.9%) | |
| Deficiency: <20 ng mL $^{-1}$ | | 44 (88%) | | |
| Mean values of vit | amin D, TSH, T3 and T4 | | | |
| Vitamin D | | 18.8 <u>+</u> 10.8 ng mL ⁻¹ | | |
| TSH | | 13 <u>+</u> 11.1 | | |
| Т3 | | 4.3 <u>+</u> 1.6 | | |
| T4 | | 13 <u>+</u> 3.3 | | |
| Parameter 1 | Parameter 2 | r-value | p-value | |
| Correlation betwe | en vitamin D, TSH, T3 aı | nd T4 | | |
| Vitamin D | TSH | -0.281* | 0.046 | |
| Vitamin D | T3 | -0.034 | 0.811 | |
| Vitamin D | T4 | 0.355* | 0.011 | |
| TSH | Т3 | -0.081 | 0.572 | |
| TSH | T4 | -0.319* | 0.023 | |
| Т3 | T4 | -0.291* | 0.039 | |

*Correlation is significant at the 0.05 level (2-tailed)

| Table 2. Convolution | | | Inviala and DA4 |
|----------------------|---------------|-----------|------------------|
| Table 2: Correlation | between Serum | vitamin D | levels and Bivil |

| Distribution | | Number of patients (%) | | |
|---------------------|----------------------|------------------------|---------|--|
| BMI Characteristics | 5 | | | |
| Normal | | 10 (19.6%) | | |
| Overweight | | 5 (9.8%) | | |
| Obese | | 36 (70.6%) | | |
| Parameter 1 | Parameter 2 | r-value | p-value | |
| Correlation betwee | en Vitamin D and BMI | | | |
| Vitamin D | BMI | 0.272 | 0.054 | |
| Vitamin D | BMI | 0.272 | 0.054 | |

*Correlation is significant at the 0.05 level (2-tailed)

Correlation between serum Vitamin D and BMI: Fifty-one patients were categorized into the BMI classifications. The participants BMIs were classified as normal (10 and 19.6%), overweight (5 and 9.8%) or obese (36 and 70.6%). However, the correlation between vitamin D and BMI was statistically non-significant, with r-value of 0.272 (Table 2).

DISCUSSION

The results indicated a strong association with low levels of vitamin D and higher TSH levels in up to 90% of the participants with statistical significance. The levels of T3 and T4 were in the normal range, which conformed to the presentation of autoimmune hypothyroidism¹⁷. Out of 51 patients, 36 patients were obese and had a deficiency in vitamin D; or however, the correlation between vitamin D and BMI had no statistical significance. Vitamin D deficiency is quite prevalent in adult population of Saudi Arabia, with a higher incidence in females¹⁸. The link between low vitamin D and autoimmune hypothyroidism is still under debate, some denying any role and others suggested a strong association¹⁹⁻²¹. The present study has validated the findings of previous studies which supported a role of low vitamin D levels and occurrence of autoimmune thyroid disease. A study conducted on 208 patients in Japan observed serum vitamin D deficiency (value below 25 nmoL L⁻¹) was found in 40% of female patients and in 18% of male patients (p<0.005)²². Another study conducted on autoimmune hypothyroid patients in Greece revealed that more than 85% had low levels of vitamin D as well as high levels of anti-thyroid antibodies²³. Further on, supplementation with vitamin D in these patients for 4 months (cholecalciferol, CF) resulted in a significant decrease (20.3%) in serum anti-thyroid peroxidase, which is an important marker of autoimmunity. Similar to the present study, Mackawy et al.14 indicated a strong association between hypovitaminosis D and hypothyroidism. The findings included significantly lower serum vitamin D 25 (OH) levels in hypothyroid patients than in controls (t = -11.128, p = 0.000). In addition, Kivity et al.24 confirmed higher titre of autoantibodies in patients who presented with low vitamin D levels. The study also indicated higher prevalence of vitamin D deficiency in patients with AITDs compared with healthy individuals (72% versus 30.6%; p<0.001). A similar study was also conducted on three types of autoimmune thyroid disease, including Grave's disease (GD), Hashimoto's thyroiditis (HT) and post-partum thyroiditis (PPT). The levels of vitamin D were significantly low in patients with AITD (GD and HT) when compared with healthy control²⁵. Studies have shown an important role of vitamin D in regulation of immune

system, VDR receptors have been found on cells linked to immunity (macrophages, dendritic cells. T and B Cells)²⁶. An important effect of vitamin D is the suppression of dendritic cells, which function mainly in production of autoimmunity and destruction of self-tolerance²⁷.

The association between vitamin D deficiency and obesity is still not very clear. Thus far, the quality of studies and the methods used have not been very conclusive. In a study carried out in Norway, Vitamin D and calcium levels were found to be deficient in obese men and women²⁸. It is possible that obese individuals expose less skin to the sun less often than non-obese individuals, resulting in reduced synthesis of vitamin D. BMI, body fat % and sunbathing have been shown to be related in a population-based sample, although another study found no relationship in individuals aged over 65 years^{29, 30}.

A variety of mechanisms by which vitamin D may influence adiposity and energy balance have been proposed, but interventional studies have thus far been inconclusive at least in part due to methodological issues ³¹. In summary, the data pertaining to this topic is limited and larger scale prospective studies are required. These findings further supported the association of low vitamin D and occurrence of autoimmune disorders. The present study not only supported the existing data on association of hypovitaminosis and AITD, but also further validates the findings by providing significant statistical correlation between low serum vitamin D levels and TSH. According to the results of this study, the recommendation is to evaluate and replace vitamin D levels in all patients presenting with autoimmune hypothyroid disease.

CONCLUSION

The current study aimed to evaluate the correlation between vitamin D and autoimmune hypothyroidism, as well as the correlation between vitamin D and BMI. The majority 90% of the female patients with newly diagnosed autoimmune hypothyroidism had low vitamin D levels. The diagnosis of autoimmune hypothyroid was confirmed with high levels of TSH, normal level T3 and T4 and testing positive for ATA. On the other hand, no correlation was noted between BMI and vitamin D levels.

SIGNIFICANCE STATEMENT

This study has validated the association between low vitamin D levels and the occurrence of autoimmune hypothyroidism. This study can be beneficial in terms of

planning recommendations for the treatment of autoimmune hypothyroid disease. Furthermore, vitamin D supplementation may reduce the duration and improve the recovery of patients suffering from this disease.

ACKNOWLEDGMENTS

The authors would like to express their gratitude to the Deanship of the College of Pharmacy, King Khalid University. The authors extend their appreciation to the Scientific Research Deanship at King Khalid University for supporting this work through General Research Project funds under Grant number (G.R.P- 227-39).

REFERENCES

- Madariaga, A.G., S.S. Palacios, F. Guillen-Grima and J.C. Galofre, 2014. The incidence and prevalence of thyroid dysfunction in Europe: A meta-analysis. J. Clin. Endocrinol. Metab., 99: 923-931.
- 2. Kato, S., 2000. The function of vitamin D receptor in vitamin D action. J. Biochem., 127: 717-722.
- Nagpal, S., S. Na and R. Rathnachalam, 2005. Noncalcemic actions of vitamin D receptor ligands. Endocr. Rev., 26: 662-687.
- 4. Arnson, Y., H. Amital and Y. Shoenfeld, 2007. Vitamin D and autoimmunity: New aetiological and therapeutic considerations. Ann. Rheum. Dis., 66: 1137-1142.
- Taylor, P.N., D. Albrecht, A. Scholz, G. Gutierrez-Buey, J.H. Lazarus, C.M. Dayan and O.E. Okosieme, 2018. Global epidemiology of hyperthyroidism and hypothyroidism. Nat. Rev. Endocrinol., 14: 301-316.
- 6. Bikle, D., 2009. Nonclassic actions of vitamin D. J. Clin. Endocrinol. Metab., 94: 26-34.
- Townsend, K., K.N. Evans, M.J. Campbell, K.W. Colston, J.S. Adams and M. Hewison, 2005. Biological actions of extrarenal 25-hydroxyvitamin D-1α-hydroxylase and implications for chemoprevention and treatment. J. Steroid Biochem. Mol. Biol., 97: 103-109.
- Smolders, J., K.G. Schuurman, M.E.V. Strien, J. Melief and D. Hendrickx *et al.*, 2013. Expression of vitamin D receptor and metabolizing enzymes in multiple sclerosis-affected brain tissue. J. Neuropathol. Exp. Neurol., 72: 91-105.
- 9. Mathieu, C., C. Gysemans, A. Giulietti and R. Bouillon, 2005. Vitamin D and diabetes. Diabetologia, 48: 1247-1257.
- Schneider, L., A.S.P. dos Santos, M. Santos, R.M. da Silva Chakr and O.A. Monticielo, 2014. Vitamin D and systemic lupus erythematosus: State of the art. Clin. Rheumatol., 33: 1033-1038.

- 11. Antico, A., M. Tampoia, R. Tozzoli and N. Bizzaro, 2012. Can supplementation with vitamin D reduce the risk or modify the course of autoimmune diseases? A systematic review of the literature. Autoimmunity Rev., 12: 127-136.
- Shin, D.Y., K.J. Kim, D. Kim, S. Hwang and E.J. Lee, 2014. Low serum vitamin D is associated with anti-thyroid peroxidase antibody in autoimmune thyroiditis. Yonsei Med. J., 55: 476-481.
- Bozkurt, N., B. Karbek, B. Ucan, M. Sahin, E. Cakal, M. Ozbek and T. Delibasi, 2013. The association between severity of vitamin D deficiency and Hashimoto's thyroiditis. Endocr. Pract., 19: 479-484.
- 14. Mackawy, A.M.H., B.M. Al-Ayed and B.M. Al-Rashidi, 2013. Vitamin D deficiency and its association with thyroid disease. Int. J. Health Sci., 7: 267-275.
- Anaraki, P.V., A. Aminorroaya, M. Amini, F. Momeni, A. Feizi, B. Iraj and A. Tabatabaei, 2017. Effect of vitamin D deficiency treatment on thyroid function and autoimmunity markers in Hashimoto's thyroiditis: A double-blind randomized placebocontrolled clinical trial. J. Res. Med. Sci., Vol. 22. 10.4103/jrms.JRMS_1048_16
- Sindi, H.A., F.M. Yousef and B.A. Aljahdali, 2018. Vitamin D status and autoimmune disease (Hashimoto's thyroiditis) in Saudi Arabian women. Int. J. Pharmaceut. Phytopharmacol. Res., 8: 21-26.
- 17. Sanyal, D., 2014. Spectrum of Hashimoto's thyroiditis: Clinical, biochemical and cytomorphologic profile. Indian J. Med. Res., 140: 710-712.
- Alsuwaida, A.O., Y.M. Farag, A.A. Al Sayyari, D.H. Mousa and F.F. Alhejaili *et al.*, 2013. Prevalence of vitamin D deficiency in Saudi adults. Saudi Med. J., 34: 814-818.
- Musa, I.R., G.I. Gasim, S. Khan, I.A. Ibrahim, H. Abo-alazm and I. Adam, 2017. No association between 25 (OH) vitamin D level and hypothyroidism among females. Open Access Maced. J. Med. Sci., 5: 126-130.
- 20. Yasmeh, J., F. Farpour, V. Rizzo, S. Kheradnam and I. Sachmechi, 2016. Hashimoto thyroiditis not associated with vitamin D deficiency. Endocr. Pract., 22: 809-813.
- 21. Effraimidis, G., K. Badenhoop, J.G. Tijssen and W.M. Wiersinga, 2012. Vitamin D deficiency is not associated with early stages of thyroid autoimmunity. Eur. J. Endocrinol., 167: 43-48.
- 22. Yamashita, H., S. Noguchi, K. Takatsu, E. Koike and T. Murakami *et al.*, 2001. High prevalence of vitamin D deficiency in Japanese female patients with Graves' disease. Endocr. J., 48: 63-69.
- Mazokopakis, E.E., M.G. Papadomanolaki, K.C. Tsekouras, A.D. Evangelopoulos, D.A. Kotsiris and A.A. Tzortzinis, 2015. Is vitamin D related to pathogenesis and treatment of Hashimoto's thyroiditis? Hell J. Nucl. Med., 18: 222-227.
- 24. Kivity, S., N. Agmon-Levin, M. Zisappl, Y. Shapira and E.V. Nagy *et al.*, 2011. Vitamin D and autoimmune thyroid diseases. Cell. Mol. Immunol., 8: 243-247.

- Ma, J., D. Wu, C. Li, C. Fan and N. Chao *et al.*, 2015. Lower serum 25-hydroxyvitamin D level is associated with 3 types of autoimmune thyroid diseases. Medicine, Vol. 94. 10.1097/MD.00000000001639
- 26. Cantorna, M.T., L. Snyder, Y.D. Lin and Y. Yang, 2015. Vitamin D and 1,25(OH)₂D regulation of T cells. Nutrients, 7: 3011-3021.
- 27. Bscheider, M and E.C. Butcher, 2016. Vitamin D immunoregulation through dendritic cells. Immunology, 148: 227-236.
- O'Neill, C., A. Kazantzidis, M. Ryan, N. Barber and C. Sempos *et al.*, 2016. Seasonal changes in vitamin Deffective UVB availability in Europe and associations with population serum 25-hydroxyvitamin D. Nutrients, Vol. 8, No. 9. 10.3390/nu8090533

- 29. Kull, M., R. Kallikorm and M. Lember, 2009. Body mass index determines sunbathing habits: Implications on vitamin D levels. Internal Med. J., 39: 256-258.
- 30. Harris, S.S and B. Dawson-Hughes, 2007. Reduced sun exposure does not explain the inverse association of 25hydroxyvitamin D with percent body fat in older adults. J. Clin. Endocrinol. Metab., 92: 3155-3157.
- 31. Song, Q. and I.N. Sergeev, 2012. Calcium and vitamin D in obesity. Nutr. Res. Rev., 25: 130-141.